

GOPEN ACCESS

Citation: Croucher NJ, Kagedan L, Thompson CM, Parkhill J, Bentley SD, Finkelstein JA, et al. (2015) Selective and Genetic Constraints on Pneumococcal Serotype Switching. PLoS Genet 11(3): e1005095. doi:10.1371/journal.pgen.1005095

Editor: Diarmaid Hughes, Uppsala University, Sweden

Received: November 8, 2014

Accepted: February 23, 2015

Published: March 31, 2015

Copyright: © 2015 Croucher et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Sequence data have been deposited in the European Nucleotide Archive (ENA) under project ERP000809.

Funding: NJC was supported by an AXA Foundation post-doctoral fellowship and a Sir Henry Dale Fellowship jointly funded by the Wellcome Trust and Royal Society (Grant Number 104169/Z/14/Z). This work was funded by the National Institute of Allergy and Infectious Diseases of the US National Institutes of Health award R01Al066304 and Wellcome Trust grant 098051. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. **RESEARCH ARTICLE**

Selective and Genetic Constraints on Pneumococcal Serotype Switching

Nicholas J. Croucher^{1,2*}, Lisa Kagedan³, Claudette M. Thompson³, Julian Parkhill⁴, Stephen D. Bentley⁴, Jonathan A. Finkelstein^{5,6}, Marc Lipsitch^{2,3}, William P. Hanage²

 Department of Infectious Disease Epidemiology, Imperial College London, London, United Kingdom,
Center for Communicable Disease Dynamics, Department of Epidemiology, Harvard T. H. Chan School of Public Health, Boston, Massachusetts, United States of America, 3 Department of Epidemiology and Department of Immunology and Infectious Diseases, Harvard T. H. Chan School of Public Health, Boston, Massachusetts, United States of America, 4 Pathogen Genomics, The Wellcome Trust Sanger Institute, Wellcome Trust Genome Campus, Hinxton, Cambridge, United Kingdom, 5 Department of Population Medicine, Harvard Medical School and Harvard Pilgrim Health Care Institute, Boston, Massachusetts, United States of America, 6 Division of General Pediatrics, Boston Children's Hospital, Boston, Massachusetts, United States of America

* n.croucher@imperial.ac.uk

Abstract

Streptococcus pneumoniae isolates typically express one of over 90 immunologically distinguishable polysaccharide capsules (serotypes), which can be classified into "serogroups" based on cross-reactivity with certain antibodies. Pneumococci can alter their serotype through recombinations affecting the capsule polysaccharide synthesis (cps) locus. Twenty such "serotype switching" events were fully characterised using a collection of 616 whole genome sequences from systematic surveys of pneumococcal carriage. Eleven of these were within-serogroup switches, representing a highly significant (p < 0.0001) enrichment based on the observed serotype distribution. Whereas the recombinations resulting in between-serogroup switches all spanned the entire cps locus, some of those that caused within-serogroup switches did not. However, higher rates of within-serogroup switching could not be fully explained by either more frequent, shorter recombinations, nor by genetic linkage to genes involved in β -lactam resistance. This suggested the observed pattern was a consequence of selection for preserving serogroup. Phenotyping of strains constructed to express different serotypes in common genetic backgrounds was used to test whether genotypes were physiologically adapted to particular serogroups. These data were consistent with epistatic interactions between the cps locus and the rest of the genome that were specific to serotype, but not serogroup, meaning they were unlikely to account for the observed distribution of capsule types. Exclusion of these genetic and physiological hypotheses suggested future work should focus on alternative mechanisms, such as host immunity spanning multiple serotypes within the same serogroup, which might explain the observed pattern.

Competing Interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: WPH has consulted for GlaxoSmithKline. ML has consulted for Pfizer and Novartis.

Author Summary

Streptococcus pneumoniae is a major respiratory pathogen responsible for a high burden of morbidity and mortality worldwide. Current anti-pneumococcal vaccines target the bacterium's polysaccharide capsule, of which at least 95 different variants ('serotypes') are known, which are classified into 'serogroups'. Bacteria can change their serotype through genetic recombination, termed 'switching', which can allow strains to evade vaccine-induced immunity. By combining epidemiological data with whole genome sequencing, this work finds a robust and unexpected pattern of serotype switching in a sample of bacteria collected following the introduction of routine anti-pneumococcal vaccination: switching was much more likely to exchange one serotype for another within the same serogroup than expected by chance. Several hypotheses are presented and tested to explain this pattern, including limitations of genetic recombination, interactions between the genes that determine serotype and the rest of the genome, and the constraints imposed by bacterial metabolism. This provides novel information on the evolution of *S. pneumoniae*, particularly regarding how the bacterium might diversify as newer vaccines are introduced.

Introduction

Streptococcus pneumoniae is a human nasopharyngeal commensal bacterium and important respiratory pathogen. The ability of the pneumococcus to cause invasive disease appears critically dependent upon its polysaccharide capsule [1], of which at least 95 immunologically-distinguishable capsular variants (serotypes) are known [2–7]. This structure inhibits the recognition of subcapsular protein antigens by the adaptive immune system and the binding of phosphorylcholine residues by C-reactive protein, thereby reducing the rate of complement deposition on the bacterial surface [8]. Children develop anticapsular antibodies after exposure to the bacterium, although evidence for their impact on disease risk is mixed [9]. Data indicating this immune response provides protection against nasopharyngeal carriage has been found for only a few serotypes [10–12], though mathematical models suggest subtle effects on carriage may exist and help maintain the high level diversity of serotypes observed in pneumococcal populations [13].

In contrast to natural immunity, that induced by the seven-valent protein conjugate polysaccharide vaccine (PCV7) reduces acquisition of vaccine serotypes in the nasopharynx by 50% [14] or more [15]. Following the introduction of PCV7, a substantial fall in the carriage of the seven vaccine serotypes (4, 6B, 9V, 14, 18C, 19F and 23F) was observed without any substantial overall reduction in the rates of pneumococcal colonisation [16,17]. This was primarily the result of an increase in the rate of carriage of non-vaccine type strains, termed 'serotype replacement' [18–20]. In some cases non-vaccine type strains are closely related to vaccine type strains from which they have been derived by 'serotype switching' [21]. In these cases, strains have altered their serotype through the acquisition of a different capsular polysaccharide synthesis (*cps*) locus via genetic transformation.

Many serotypes, though distinguishable by certain antisera (called "factor sera"), nonetheless may be clustered into "serogroups" based on cross-reactivity with other antisera; these groups have often been found to correspond to sets of similar polysaccharide structures [2]. It was originally hoped that PCV7 would provide cross-protection against 'vaccine-related' serotypes: those within the same serogroups as a serotype included in the vaccine [22]. However, the only case in which such an effect was observed was the protection against colonisation with serotype 6A resulting from the inclusion of serotype 6B in PCV7; several other vaccine-related types increased in prevalence post-PCV7 [23]. Contributing to this pattern was an apparent tendency for serotype switches to exchange one serotype for another within the same serogroup more often than expected by chance: of nine switches inferred using multilocus sequence typing (MLST) in a systematic collection of carried pneumococci from Massachusetts, three were within-serogroup (p = 0.043) [24]. Hence genotypes successful prior to the introduction of PCV7 were able to persist expressing a similar capsule that was not recognised by vaccine-induced immunity. This was not a pattern expected *a priori*, as random acquisition of a new serotype from the full, diverse set of non-vaccine type capsules should usually result in a change of serogroup.

This work assesses the relative likelihood of alternative explanations of the observed pattern of serotype switching based on whole genome sequencing data from this Massachusetts-based collection [25]. The first is that the detected propensity for within-serogroup serotype switching is a consequence of the constraints of genetic transformation. Pneumococcal cps loci are typically 10–30 kb in size, usually necessitating a similarly long recombination to cause a change of serotype [26], whereas homologous recombinations have an exponential distribution of sizes with a mean length of between two and seven kilobases [27-29]. Aside from known exceptions such as serogroups 7, 17, 33 and 35, the cps loci corresponding to a single serogroup are closely related [2]. Therefore, within-serogroup serotype switching may be accomplished through relatively short, more frequent, recombination events that do not replace the entire *cps* locus [30]. Another possibility is that patterns of serotype switching may be affected by the flanking *pbp1a* and *pbp2x* genes, which are crucial in determining β -lactam resistance. As resistance is associated with a limited number of serotypes [31], and long recombinations that change serogroup could lead to the acquisition of resistance [32] or risk reversing any beneficial acquisition of resistance-associated pbp1a or pbp2x alleles [29,33], it may be that maintenance of a strain's β -lactam susceptibility (or lack thereof) affects the patterns of serotype switching.

Alternatively, rather than representing a limitation on the rate of recombination, the pattern of switching may reflect the consequence of constraints imposed by epistatic interactions with other loci in the chromosome [26]. Such limitations may reflect physiological or metabolic specialisation to production of particular capsule types; alternatively, it may be important to coordinate the expression of particular serogroups with certain alleles of immunogenic surface proteins, in order to maintain discordant sets of antigens in distinct lineages [34]. The last explanation to be considered, which also involves the host immune response, relies on the presence of natural or vaccine-induced antibodies that cross-react with all serotypes within a serogroup. This would lead to elevated rates of co-colonisation between strains of the same serogroup, as bacteria would be confined to a subpopulation of hosts that do not exhibit immunity to either of their capsules. A consequence of this would be increased genetic exchange between members of the same lineage, including an elevated rate of within-serogroup serotype switching.

Results

Association between serotypes and genotypes

Of the 616 draft genomes previously analysed [25], 491 fell into fifteen monophyletic sequence clusters (SCs) of related isolates within which serotype switching could be investigated (<u>Fig. 1</u>). These sequence clusters included representatives of 19 serotypes spread across eleven serorgroups. In seven sequence clusters, the serotype was stable across the clade, leaving eight clusters in which at least one switch had occurred.

There was no positive correlation between the number of isolates sampled within a sequence cluster and the number of serotypes it contained (<u>Fig. 2A</u>). However, a significant correlation was observed between the number of polymorphic sites, as ascertained through a lineage-





Fig 1. Prevalence of serotype switching in the overall population. The maximum likelihood phylogeny derived from a codon alignment of 'core' protein coding sequences is displayed, with fifteen monophyletic sequence clusters (SC1-SC15) labelled by arcs around the edge of the tree. Those sequence clusters in which no serotype switching was observed are marked by black arcs. Those sequence clusters in which multiple serotypes were observed are individually coloured, with the most prevalent serotype represented by light shading, and alternative serotypes indicated by darker shading or dashed lines.

specific analysis of whole genome alignments, and the number of serotypes in a cluster (Fig. 2B). When these sites were split into the number of point mutations and homologous recombination events per sequence cluster (see <u>Methods</u>), there was no significant relationship with the former, whereas the latter had the strongest correlation of any measure of genetic diversity (Fig. 2C and D). Therefore sequence clusters that exhibit serotype diversity are those



Fig 2. Associations between properties of a sequence cluster and its capsular diversity. For each of the fifteen monophyletic sequence clusters, the number of serotypes it contained was plotted against (A) the number of representatives of the sequence cluster; (B) the overall genetic diversity of the sequence cluster, as represented by the number of polymorphic sites identified from whole genome alignments; (C) the number of point mutations reconstructed as having occurred over the history of the sequence cluster; and (D) the number of homologous recombination events reconstructed as having occurred over the history of the sequence cluster.

that have experienced more recombination events throughout their genome over their observable evolutionary history, either as a consequence of a high rate of recombination over their recent diversification, or lower recombination rates over a longer period of time.

Frequent within-serogroup switching

Across the species, the 95 currently known S. pneumoniae serotypes are divided into 48 serogroups such that 2.2% of comparisons between different serotypes are serogroup concordant. However, many serotypes are rarely observed; the 32 serotypes observed across all 616 isolates from Massachusetts were distributed such that 3.0% of serotype comparisons were serogroup concordant. Seventeen serotypes were observed within the eight sequence clusters that appear to contain examples of serotype switching; 7.2% of all pairwise comparisons between these serotypes were serogroup-concordant, suggesting these isolates expressed a comparatively limited number of serogroups. Yet when only considering the subset of these comparisons where the different serotypes were found within the same sequence cluster, 29% were serogroup concordant (S1 Table), as some individual sequence clusters were associated with multiple serotypes from the same serogroup. The significance of this enrichment was assessed by a permutation test that randomly assigned serotypes to these eight sequence clusters according to the number of serotypes originally observed in the sequence cluster. This found the high level of serogroup-concordant comparisons within sequence clusters to be statistically significant (p = 0.0007 from 10,000 permutations). As sequence clusters were identified through a clustering algorithm [25], this observation is independent of any phylogenetic analyses. Therefore this test indicates isolates with similar core genomes are more likely to express capsules of the same serogroup than expected from the overall capsular diversity of the isolate collection, even when accounting for isolates sharing the same serotype through common descent.

Greater resolution can be achieved by using the phylogenetic analysis of the whole genome alignments for each of the sequence clusters in which serotype switching occurred (S1 Text & Fig. 3) [25]. These attempt to reconstruct the history of the lineages more accurately by identifying recombination events and excluding the horizontally acquired polymorphisms they introduce from the point mutations used to generate the tree [29]. These reconstructions found that some within-serogroup switching occurred multiple times in parallel in the same sequence cluster: for instance, in SC9 the ancestral type of 23A was replaced by 23B on three occasions, and 23F on three occasions. Similarly within SC13, the ancestral 6A type is exchanged for 6C on two occasions in parallel. In both cases, these examples of convergent evolution are independently supported by the divergence between the *cps* loci, encoding the same serotype, imported by separate capsule switching events (S1 Fig). A further switch to 6C, this time from 6B, was observed within SC6. However, the other changes within this sequence cluster cannot be reliably inferred as they occur on long branches that are difficult to reconstruct. These 'missing data' include at least two between-serogroup switches, but also likely within-serogroup switches given that the cluster includes 23A, 23B and 23F isolates.

By contrast, each of the robustly inferred between-serogroup switches occurred only once. Hence this more detailed analysis revealed an even greater tendency to swap a serotype for one within the same serogroup than the previous results. Overall, eleven of the twenty serotype switches (55%) fully defined in this population did not result in a change in serogroup. A further permutation test, in which the derived serotypes were randomly assigned to each ancestral serotype through resampling without replacement 10,000 times (see Methods), found this result to be highly significant (p < 0.0001). Therefore this collection provides robust evidence that within-serogroup switching is more common than expected by chance, relative to the rate of between-serogroup switching.



Fig 3. Serotype switching events within sequence clusters. The evolution of each of the monophyletic sequence clusters highlighted in Fig. 1 was reconstructed from a whole genome alignment (see <u>Methods</u>). The resulting phylogenies are annotated with the serotype of the isolates. Stars on the branches of each phylogeny mark instances where putative recombinations affected the *cps* locus. These stars are coloured according to the inferred consequences of the recombination. Where these sequence imports led to a change in serotype, this is annotated: in red for within-serogroup switches, in blue for between-serogroup switches, and in green, where the ancestral serotype cannot be reliably determined.

Length of serotype switching recombinations

The first hypothesis that might explain this observation concerns the mechanics of recombination: the smaller recombinations necessary to alter a *cps* locus to a related sequence generating a similar serotype may be more frequent than the replacement of the entire locus. The withinserogroup switches observed in this collection all require only minor changes in the *cps* locus: the serogroup 23 *cps* loci are thought only to differ in the sequence of their *wzy* polymerase, as is also the case for serotypes 19A and 19F [35]. Similarly, switching between serotypes 6A and 6B can occur through a single polymorphism in the *wciP* gene [30], while 6A and 6C only differ in polymorphisms in their *wciN* gene [35].

Including only the switch from serotype 6B to 6C from SC6 (see S1 Text), the analysis of whole genome alignments used to generate Fig. 3 identified 50 putative homologous recombination events that overlapped the cps loci of the relevant reference genome sequences (Fig. 4). The nine recombinations that resulted in a change in serogroup had a median length of 42.7 kb, and all spanned the entire *cps* locus. The eleven recombinations that led to a within-serogroup switch had a shorter median length, at 30.4 kb (Fig. 5A). These almost all spanned the 5' region of the *cps* locus, but in some cases did not extend so far as the 3' end. In two cases, this could be ascribed to the presence of the *rml* rhamnose synthesis operon at the 3' end of the *cps* locus. Both the acquisition of the 6B capsule in SC13, and the 6C capsule in SC6, terminated within this gene cluster; however, the *rml* operon is found in several serogroups [2], and recombinations causing between-serogroup switches have previously been observed to end within it [29]. Only three recombination events were observed where the 3' recombination breakpoint occurred in a region that might be considered serogroup-specific. Two of these were switches from 23A to 23F within SC9 (Fig. 4). In both cases, the wzy polymerase gene that distinguished these *cps* loci was replaced; however, the recombinations were far more extensive than this minimal alteration, as they extended to or beyond the 5' boundary of the cps locus. The switch to 6C within SC13 that ended before the 3' boundary of the locus was also far more extensive than simply encompassing the *wciN* gene. The overall difference in the distribution of lengths between recombinations causing within- and between-serogroup switches was not significant (Fig. 5; Wilcoxon rank sum test, W = 35, p value = 0.29). By contrast, the 30 recombinations inferred to overlap with the *cps* loci that did not affect serotype had a median length of 11.7 kb and were significantly shorter than both the recombinations that alter serogroup (Wilcoxon rank sum test, W = 22, p value = 3.4×10^{-5}) and those that caused within-serogroup switching (Wilcoxon rank sum test, W = 61, p value = 0.0016). As the majority of within-serogroup switches were caused by recombinations long enough to cause changes in serogroup, restrictions on transformation event length cannot fully explain the observed pattern of switching.

There was extensive variation in the boundaries of the three classes of recombination event. For both 5' and 3' breakpoints outside the *cps* locus, the distance between the breakpoint and the *cps* locus edge (Fig. 4B) followed an approximately exponential distribution, as previously observed for experimental transformants at the *cps* locus [28]. However, these observed recombinations were longer than the length of transformation events observed experimentally, with a rate of decay approximately an order of magnitude lower in this work: the rate constant for the decline on the left side was 8.07×10^{-5} bp⁻¹ (95% confidence interval: $8.06 \times 10^{-5} - 8.09 \times 10^{-5}$ bp⁻¹), while on the right side it was 6.75×10^{-5} bp⁻¹ (95% confidence interval: $6.74 \times 10^{-5} - 6.77 \times 10^{-5}$ bp⁻¹), as opposed to the previous estimates of ~ 3.4×10^{-4} bp⁻¹. This is likely partly a consequence of the events described in Fig. 4 being composed long "mosaics" of recombinant DNA segments [28], and also potentially representing larger "macrorecombination" events [36].

The *pbp2x* and *pbp1a* genes, crucial determinants of β -lactam resistance, are found 9–10 kb upstream and downstream of the *cps* locus respectively. Seven of the recombinations shown in



Fig 4. Extent of recombinations affecting the *cps* **locus.** (A) Positions of recombinations affecting the *cps* locus. Each of the recombinations identified in Fig. 3 is represented by a horizontal bar coloured according to the inferred impact of the recombination. Recombinations causing a change in serotype are uniquely annotated to correspond with Fig. 3. The grey column represents the extent of the *cps* locus; the width displayed is that of the longest *cps* locus across the different sequence clusters, and each recombination is scaled relative to the positions of this *cps* locus' boundaries. (B) Extent of recombinations affecting the *cps* locus. The degree to which the recombinations displayed in (A) impinge on the flanking regions is summarised, with the dashed line representing the falling numbers of recombinations extending to bases further removed from the *cps* locus edges (displayed on the same scale as (A)). The red lines represent the fit of exponential decay curves to these trends. The tick marks above the plots represent 10 kb intervals.

PLOS GENETICS





Fig 5. Properties of recombinations affecting the *cps* **locus.** In each boxplot, the recombinations affecting the *cps* locus are grouped according to their inferred impact on serotype. The first three graphs describe the recombinations affecting the *cps* locus. The values represent (A) overall length of recombination, (B) position of the 5' edge of the recombination relative to the 5' edge of the *cps* locus, and (C) position of the 3' edge of the recombination relative to the 3' edge of the *cps* locus. The boxplot in (D) represents the total length of genome-wide recombinations on the same phylogenetic branch as the serotype switching events.

Fig. 4 affected both, all of which caused a change of serotype. However, they were split between within- and between-serogroup switches, and none significantly affected the strain's β -lactam resistance (S2 Table & S2 Text). Hence serotype switches did not appear to be driven by selection for the acquisition of β -lactam resistance. Nevertheless, it is possible that maintenance of a genotype's β -lactam resistance level may counterselect against recombination events that substantially alter *pbp1a* or *pbp2x* [33].

Adaptation between serotype and genotype

If the distribution of capsular variation does not reflect the properties of genetic transformation, then selection may be important in driving the observed pattern. Possible selective pressures include adaptive immune responses in the host population [34], or physiological constraints relating to the bacterium. To test the latter possibility, capsule-switched variants were constructed in common genetic backgrounds and characterised through recording growth curves.

The association of SC9 with serogroup 23 was investigated by knocking out the native cps locus of isolate R34-3029, and using this genetic background to construct one mutant in which the native 23F capsule type was restored, a second through a within-serogroup switch to capsule type 23B, and two further mutants through between-serogroup switches to 18C and 6B. Capsule type 18C was selected as it was the only non-serogroup 23 capsule type observed within SC9, whereas 6B was not found within SC9 but has similar properties to 23F [37]. Both serogroup 23 mutants were found to grow similarly quickly (Fig. 6A) and substantially faster than the two mutants bearing capsules of other serogroups. This suggested that these bacteria may be adapted to serogroup 23 capsules. However, comparisons of the original isolates used as the sources of the cps loci (S2 Fig) indicated that the apparent reduction in fitness of SC9 mutants expressing non-serogroup 23 capsule types might reflect an intrinsically slower growth phenotype associated with these capsule types. Furthermore, when the reciprocal exchange of cps loci between isolates BR1014 and R34-3029 was performed, BR1014 grew faster following restoration of its native 23B capsule relative to the variant into which the 23F capsule had been introduced (Fig. 6A). Therefore changes in capsule type were generally associated with a greater reduction in growth rate than reinstatement of the original capsule type; however, this inhibition could be observed regardless of whether the new serotype was of a different serogroup or not.

To test this hypothesis further, six additional mutants were constructed using two SC13 isolates in which the *cps* locus had been knocked out: MD5037, originally of 6B, and MD5030, originally of 6C (Fig. 6B). The 6B, 6C and 15F capsule types were introduced into both. This comparison allowed a more precise test for interactions between the *cps* locus and the rest of the genome than the experiments with SC9 isolates, as SC13 isolates exhibited more consistent growth patterns, and the selected donor of the 15F capsule had a similar growth curve to the recipient isolates (S2 Fig). Consistent with the observations of SC9, the mutants with the restored native capsule grew fastest. However, both the within-serogroup and between-serogroup switched variants exhibited similarly inhibited growth rates. Hence all three capsule types replicated optimally in their native backgrounds, and more slowly when introduced into a non-native background. Therefore the only epistasis between the *cps* locus and the rest of the pneumococcal genome that could be inferred from these data was specific to serotype, not serogroup, and no evidence was found that the observed predominance of within-serogroup switches is explained by serogroup-specific adaptations.

Potential for epistatic interactions

It remains possible that epistasis affects other phenotypes that are not easily assayed in the laboratory; it may be possible to infer these from the genomic data. One factor likely to impact on



Fig 6. Growth curves comparing the fitness of genetic backgrounds expressing different serotypes. (A) Effect of capsule switching on the growth of two isolates from SC9. Three replicate experiments (i-iii) are shown in which six mutants were compared. Four were constructed based on isolate R34-3029 (diamond points): one in which the native 23F capsule was restored (orange); a within-serogroup switch to the 23B capsule (red); a between-serogroup switch to the 18C capsule (light blue); and a further between-serogroup switch to the 6B capsule (blue). Two other mutants were constructed based on isolate BR1014 (square points): one in which the native 23B capsule was restored (orange), and a within-serogroup switch to 23F (red). (B) Effect of capsule switching on the growth of two isolates from SC13. Three replicate experiments (i-iii) are shown in which six mutants were compared. Three mutants were based on isolate MD5037 (circular points): a restoration of the native 6B capsule (orange); a within-serogroup switch to the 6C capsule (red); and a between-serogroup switch to the 15F capsule (blue). The other three mutants were based on isolate MD5030 (triangular points): a restoration of the native 6C capsule (orange), a within-serogroup switch to the 15F capsule (blue).

PLOS GENETICS

the range of capsules a strain may successfully express is its complement of carbohydrate transporters [38,39], many of which have been functionally characterised [38] and are only present in a subset of the population (S3 Fig & S3 Table). While capsules in different serogroups usually have distinct chemical compositions, capsules within the same serogroup often consist of the same moieties connected by different bonds. Hence strains may only be adapted to importing and processing certain carbohydrates at high enough rates to sustain capsule production, thereby inhibiting the acquisition of a capsule type with a divergent chemical composition. However, the distribution of transporters across the population does not provide an obvious explanation for the stable association between sequence clusters and serogroups (S3 Fig). SCs 9, 13 and 15 did not appear to have a smaller number of carbohydrate transporters than those in which a change of serogroup had occurred, nor relative to those not having changed serotype. The associated serogroups can be distinguished by the presence of ribitol (serogroup 6), *N*-acetylmannosamine (serogroup 19) and glycerol (serogroup 23). Ribitol is synthesised as part of teichoic acid common to all pneumococci; glycerol is imported by GlpF, encoded by a gene ubiquitous across the sample; and *N*-acetylmannosamine is known to be a substrate for four transporters that are common to all isolates, and one that is absent from SC15. These data do not support the hypothesis that the observed pattern of switching is the result of a constricted ability to acquire the requisite carbohydrate molecules, although epistatic interactions with other aspects of carbohydrate metabolism cannot be excluded.

Yet this analysis only uses experimentally characterised loci, which cannot encompass novel systems that may be rare in the population. It is possible in principle to identify epistatic interactions between any loci from sequence data [40]. In bacteria, interacting accessory genome loci should be present within the same genomes more often than expected by chance, after accounting for linkage and clonal structure. Three sequence clusters, each essentially an independent genotype [41], were universally of serogroup 6 (SC10, SC13 and SC14). A search for clusters of orthologous genes (COGs) that were ubiquitous in these sequence clusters but absent from sequence clusters with no serogroup 6 representatives identified three COGs, all of which corresponded to genes within the serogroup 6 cps loci. This failure to identify serogroup 6 associated genes outside the cps locus is in keeping with the experimental data that suggests adaptation is specific to a serotype, not a serogroup. Testing for association with serotype requires that the same capsule be acquired multiple times in parallel, as in SC9. Each triallelic polymorphic site in SC9 was tested for association of an allele with one of the three serogroup 23 serotypes. Twelve such sites were identified, all of which were within the cps locus. Additionally, no COGs in the accessory genome associated perfectly with any of these individual serotypes. Therefore these simple analyses provided no evidence of sites outside the cps locus that might epistatically interact with the expressed capsule type.

A more generic strategy for identifying potential epistatic interactions was not to look for associations of specific sequences with particular cps loci, but instead quantify the overall extent of changes accompanying serotype switches. If one or more non-cps loci improved the fitness of a novel serotype variant, then it might be expected that serotype switches would be associated with elevated rates of change across the rest of the genome. In order to control for the general heterogeneity in the rates of pneumococcal recombination [36], the total lengths of the recombinations occurring on the same phylogenetic branch as serotype switches (both withinand between-serogroups) were compared with those occurring on branches on which recombinations affected the cps locus but did not cause a change in serotype (Fig. 5D). This found that similar levels of contemporaneous (or near-contemporaneous) recombination occurred whether or not there was a change in serotype or serogroup. Additionally, all three types of recombination affecting the cps locus were typically associated with sufficient numbers of base substitutions being imported by recombination (S4 Fig) to saturate the mismatch repair system [42]. Hence, there was no significant evidence of non-cps loci being exchanged to facilitate the acquisition of a new serotype or serogroup, nor of mismatch repair having an impact on the patterns of serotype switching.

Discussion

Previous study of this collection of carried pneumococci, based on serotyping and MLST data, identified a borderline-significant tendency for serotype switches to occur within serogroups [21]. The availability of whole genome-based phylogenies increases the number of switches that can be investigated within the same set of isolates through identification of two classes of switch invisible to the MLST-based method: changes in serotype along a lineage in which the multilocus sequence type has also changed (e.g. SC9), and also multiple independent switches within the same sequence type (e.g. in SC13, two switches to 6C within MLST 473). This

increases the sample size of identified serotype switching events from nine to twenty. Also, these data provide more reliable phylogenetic information on the ancestral serotype: previously, this was based on the temporal order in which combinations of serotype and genotype were observed, which can be misleading in certain cases, such as SC9 [25].

Future population genomic datasets will determine how general the preponderance of within-serogroup switching is across the pneumococcal species; a recently published dataset of over 3,000 genomes systematically sampled from a Thai refugee camp, in which hosts were unvaccinated, provides an opportunity to test the reproducibility of the pattern [43]. Applying the simple permutation test to the distribution of serotypes between sequence clusters in which switching had occurred, as for the dataset described in this study, revealed the ratio of serogroup concordant to serotype-discordant comparisons to be 0.14 (p = 0.0006 from 10,000 permutations). The serotypes in this sample are markedly different from those in the Massachusetts dataset; hence this enrichment of within-serogroup switching provides independent support for the original observation. Additionally, although only one within-serogroup switch was observed in SC15 in this dataset, the acquisition of the 19A capsule in place of the 19F type has occurred at least three further times in related isolates [33].

Multiple hypotheses were proposed as potential explanations for the observed frequency of within-serogroup switches. The first was that the pattern was caused by the constraints of genetic transformation, with short recombination events resulting in partial alteration of the *cps* locus being more frequent. The suggestion that most recombinations are too short to be likely to cause a change in serotype was supported by the large number of 'silent' recombinations affecting the *cps* locus, which did not change the antigenic profile of the bacterium. While some cases are likely to reflect exchanges of sequence between distinct isolates of the same serotype, many of these events affect the 5' or 3' boundaries of the *cps* locus that are the most strongly conserved across different *cps* loci, and therefore may originate in donors with a very different *cps* locus to that of the recipient. As these exchanges do not result in antigenic changes, it may be that they represent neutral diversification, or serve a role in repairing deleterious mutations. This potential for recombination between serotypes that does not cause switching should be borne in mind when designing sequence-based serotyping methodologies that do not directly target the polymorphisms within *cps* loci that cause differences in capsule type.

Nevertheless, recombinations causing within-serogroup switching sometimes did not span the entire cps locus, as all the between-serogroup switches did. In three cases, the recombinations were sufficiently restricted in their extent so as to end within a serogroup-specific part of the cps locus; however, these were still substantially larger than the putative minimal genetic changes that could cause the same alteration of serotype. Furthermore, the enrichment for within-serogroup switching remained highly significant even when the events that did not span the entire cps locus were excluded; although this does not rule out this mechanism contributing to the observed pattern of switching, this result demonstrates it is not the sole reason underlying it. Finally, the within-serogroup switches did not appear to represent a 'neutral' set of typical recombination lengths, as they were still significantly larger than those which caused no alteration of serotype overall, although there may be a limit on the length of such recombinations before they are likely to stop being 'silent'. One possibility not considered here, subtly different to a dependence on the absolute length of a recombination, is that restriction modification systems could be limiting the transfer of cps loci. Although restriction endonucleases typically do not affect transformation, as DNA is imported in a single stranded form, in cases where genomic islands are imported, the synthesis of a complementary strand once the DNA is integrated into the genome can render the acquired locus sensitive to endonucleolysis [44]. Loci imported from a different serogroup would have greater sequence divergence, and therefore a greater set of potential target motifs that could be subject to endonucleolysis. However,

this will remain difficult to assess until our knowledge of the diverse set of pneumococcal restriction-modification systems has improved [41].

Conversely, there was no evidence that serotype switching recombinations were selected for the lengths to which they affected the regions flanking the *cps* locus and the *pbp* genes. Coincidence of serotype switches with changes in β -lactam resistance [32] was rare and there was little evidence of recombinations being constrained in the extent to which they extended into the regions flanking the cps locus. However, there are uncertainties associated with ascertaining these recombination breakpoints. One is that the evolutionary time represented by the long branches on which the changes of serogroup occur within SC1 may be sufficient to permit several rounds of transformation. This could result in overlapping recombinations being erroneously inferred to represent a single event, thereby increasing the apparent length of the recombined sequence. However, as both events affected by this bias were between-serogroup switches, this cannot account for within-serogroup switches being longer than the minimum length required for the phenotypic change. Conversely, multiple switches to the same serotype in parallel produce extensive homoplasy. Inaccurate reconstruction of such homoplasies could result in individual actual recombination events being reconstructed as multiple fragmentary events split between different branches of the phylogeny. This risks artificially shortening serotype-switching recombinations. However as this bias only affects sequence clusters in which similar *cps* loci are acquired multiple times, which also happen to be those in which withinserogroup switching is most common (SC9 and SC13), any such errors are conservative with regard to the finding that shorter recombinations alone cannot account for the observed pattern of within-serogroup switching. One potential example is switch $23A \rightarrow 23F^{**}$, as the two *cps* loci apparently imported by this recombination are not particularly closely related (S1 Fig); hence this inferred recombination may in fact represent only a partial fragment of two separate serotype switching recombinations. If this were correct, it would increase the frequency of within-serogroup switches further, and suggest that the short recombination currently annotated as the single switch $23A \rightarrow 23F^{**}$ actually corresponds to two separate, longer events that cannot be resolved with the current collection. This ambiguity highlights how unlikely it is that the numbers of parallel switching events have been overestimated as a consequence of incorrect evolutionary reconstructions. Given the substantial numbers of polymorphisms caused by capsule switching recombinations in the region of the cps locus, independent acquisitions in parallel are unlikely to be inferred without strong evidence of distinct ancestry from the rest of the chromosome. Furthermore, the diversity of the cps loci themselves, not directly used in the phylogenetic inference displayed in Fig. 3, independently support the same reconstructed pattern of switches occurring in parallel within SC9 and SC13.

As limitations to recombination do not appear to explain the observed pattern of switching, selection for certain combinations of *cps* loci and genetic backgrounds may explain the distribution of capsule across the population. Metabolic or physiological adaptation between the *cps* locus and other aspects of the pneumococcal genotype may result in a limited range of capsule types that can be successfully expressed by a given genomic background. Although serotype itself affects growth rate [45], this study found that the rate of growth *in vitro* depended on both the serotype and backbone genotype in a non-additive way. Isolates grew fastest when expressing their native capsule, although R34-3029 grew similarly fast when expressing either of the tested serogroup 23 capsule loci. However, in most cases there was no difference in the level of growth inhibition when capsules of the same, or a different, serogroup were introduced, particularly when accounting for the growth curves of the donors and recipients involved in each exchange. This is consistent with the existence of epistatic interactions between the *cps* locus and the rest of the genome, but not with the hypothesis that such interactions were serogroup-specific. It could still be that the changes required to adapt to the acquisition of a more similar

serotype are relatively small, and more likely to occur through recombination or mutation before the capsule variant is selected out of the population. Yet over the timescales easily testable in the laboratory, no evidence was found for a mechanism that would explain the frequency of within-serogroup switching.

One potential confounding problem with these experiments is the chance that non-*cps* loci could be co-transferred, and therefore influence the fitness of the resulting capsule-switched recombinants. However, previous experiments have suggested that random co-transfer of other sequence should not systematically affect the growth rates of transformants [28]. Furthermore, it might be expected that any co-transfer from the *cps* locus donor might facilitate adaptation to the acquired serotype, if there were epistatic interactions between the *cps* locus and the rest of the genome. However, the genomic data provided no evidence of natural serotype switches being associated with unusually high levels of sequence exchanges around the rest of the genome, suggesting the co-transfer of other loci is not likely to facilitate adaptation to an altered capsule type. Furthermore, there were no signs of non-*cps* loci associated with pneumococci of specific serogroups or serotypes in a manner suggestive of interactions between distinct loci in the same chromosome. Therefore, the genomic and experimental data do not support the model of simple, strong epistatic interactions determining the range of serogroups that an isolate can successfully express.

One longstanding hypothesis for the maintenance of "strain structure" in recombining pathogens is that host immunity produces strains that maintain discordant sets of antigens [34]. Here, individual capsular antigens found across serogroups might be associated with a particular set of subcapsular antigens, creating epistasis in the presence of host immunity that would not be observable in vitro. Alternatively, opportunities for recombination may be disproportionately available for strains in the same serogroup as a result of serogroup-wide immunity making hosts either susceptible or resistant to colonization with particular serogroups. Both these mechanisms rely on the assumption that anticapsular antibodies are protective against carriage, and that at least part of this protection is by responses that target epitopes shared across serotypes within a serogroup. Whether this could significantly impact the rate of genetic exchange between serotypes remains an open question. The existence of crossprotection by naturally-acquired anticapsular antibodies within a serogroup is made plausible by the observation of protection against serotype 6A from the 6B component of PCV7, and by the existence of cross-reactive antibodies used in serotyping. However, we are aware of no direct evidence for protection across a serogroup by naturally-acquired anticapsular antibodies. Furthermore, the protective immunity induced by conjugate vaccines is not serogroup-wide, as demonstrated by the post-PCV7 success of some vaccine-related serotypes, such as 19A, 23A, 23B and 6C [16].

Expanded sampling in the future may allow for these and other hypotheses to be tested further, as well as permitting more detailed searches for loci that may interact epistatically with the *cps* locus. Such information regarding which serotypes can be readily 'interchanged' may well prove important in understanding the responses to past conjugate vaccine introductions, and predicting the response to future interventions.

Methods

Permutation tests

When calculating the significance of the association of serogroups with sequence clusters, the first test resampled each of the serotypes listed in <u>S1 Table</u> without replacement until each sequence cluster had been assigned the same number of serotypes as observed in the actual dataset. This test was therefore independent of the phylogenies and reconstructed patterns of

serotype switching. Ten thousand such permutations were carried out, and in each case the calculated test statistic was the ratio of serogroup-concordant to serotype-discordant pairwise comparisons within sequence clusters. Any pairwise comparison between identical serotypes was ignored, as it would be serologically undetectable. While recombinations between isolates of the same serotype at the *cps* locus may be detected in this dataset as a subset of those events that do not alter the expressed capsule type, it is not possible to easily distinguish these from recombinations donated from isolates only of the same serogroup, or of a completely unrelated serotype, which may also not affect the recombinant's expressed capsule type, depending on how conserved different parts of the *cps* locus are across pneumococci of different serotypes. The *p* value was calculated as the proportion of permutations for which the calculated test statistic was greater than, or equal to, the value observed in the actual population (0.41). When applied to the Maela dataset, the 'secondary BAPS' groupings were used as the equivalent of sequence clusters.

The second permutation test used information from the phylogenetic reconstructions of serotype switching. The input dataset was the twenty switches for which the ancestral and derived serotypes could both be defined (Fig. 3). Ten thousand permutations were performed in which the ancestral and derived serotypes were sampled, without replacement, from those in the input dataset. In each case, the test statistic was the proportion of serotype switches that were serogroup concordant, and the *p* value calculated as the proportion of the permutations in which this statistic was equal to, or greater than, the observed value (0.55).

Reconstruction of serotype changes

The phylogenies used to identify serotype switches have been reported previously [25]. Briefly, a reference genome was assembled *de novo* for each sequence cluster, against which Illumina reads were aligned, and polymorphisms identified [28]. Sequences imported by putative recombination events were identified and removed, and a maximum likelihood phylogeny generated from the remaining clonal frame, as described [29] and validated [46] previously. The annotation of putative mobile genetic elements allowed the number of base substitutions introduced by point mutation, homologous recombination and non-homologous recombination to be estimated. All lengths and positions of recombination events are therefore relative to the reference sequences against which the Illumina reads were originally mapped. The identification of putative homologous recombination events associated with serotype switching is detailed in the supplement <u>S1 Text</u>; this also describes the assessment of within-serogroup sequence diversity, and the reconstruction of changes in serotype, achieved using the maximum likelihood approach implemented in the APE R package [47]. This approach, as well as a maximum parsimony method, was also used to reconstruct the emergence of β -lactam resistance using the trees based on the clonal frame, once each isolate had been classified as either 'resistant' (benzylpenicillin minimum inhibitory concentration $>0.06 \ \mu g \ mL^{-1}$) or 'sensitive'. The details of this analysis are also described in <u>S1 Text</u> and <u>S2 Table</u>.

Construction of in vitro serotype switches

The *cps* locus of isolates was first knocked out using a Janus cassette [48] as described previously [49]. Transformation was conducted using genomic DNA from the serotype donor and the competence stimulating peptide appropriate to the genotype. After 2 h growth, selection was performed using plates containing 500 μ g ml⁻¹ streptomycin. Colonies were replica plated to ensure loss of kanamycin resistance and expression of the altered serotype confirmed through latex agglutination (Statens Serum Institut, Copenhagen).

Comparisons of growth

Isolates were grown to mid-log phase in THY (Todd Hewitt broth containing 0.5% yeast extract), titered and frozen at -70°C in 10% glycerol. Growth was compared in parallel by inoculating 6 mL of THY with a starting concentration of 10^6 mL⁻¹ bacteria from freshly thawed frozen stock. Cultures were incubated at 37°C with 5% CO₂. At successive time points 200 µL was removed to a 96 well plate. Optical density at a wavelength of 610 nm was measured and dilutions were plated on blood agar plates to calculate live cell densities.

Supporting Information

S1 Fig. Diversity of (A) serogroup 23 *cps* loci within SC9, as compared to all serogroup 23 *cps* loci in the collection, and (B) serogroup 6 *cps* loci within SC13, as compared to all serogroup 6 *cps* loci in the collection. The *cps* loci derived from individual serotype switching events are annotated as in Fig. 3. All switches occurring in parallel imported distinct loci, indicating they are genuinely separate events. All *cps* loci derived from individual switches were monophyletic, suggesting the inferred reconstruction was correct, with the exception of switch $23A->23F^{**}$. In this case, the two *cps* loci were polyphyletic, indicating this was a single capsule switch that was followed by subsequent diversification through recombinations at the *cps* locus that did not alter serotype (as indicated by Fig. 3), or two capsule switches to 23F occurring in parallel in closely-related isolates. This latter conclusion would further strengthen the observed enrichment for within-serogroup switching. (PDF)

S2 Fig. Growth curves of isolates used as recipients and donors of *cps* loci in the capsule switching experiments. All plots show the number of viable cells at different sampling times as colony forming units per millilitre. Solid lines indicate isolates expressing the dominant serogroup in a sequence cluster; point styles match those in Fig. 6. (A) Growth curves of recipient isolates from SC9 and the donors of the serotype 6B and 18C *cps* loci. Three replicates are shown in plots (i)-(iii). (B) Growth curves of recipient isolates from SC13 and the donor of the 15F capsule type. Three replicates are again shown in plots (i)-(iii). (PDF)

S3 Fig. Distribution of carbohydrate transporters across the population. (A) Maximum likelihood phylogeny, as displayed in Fig. 1. (B) Characterised carbohydrate transporters labelled with the gene names or the corresponding locus tag codes in *S. pneumoniae* TIGR4 [EMBL accession code: AE005672] or ATCC 700669 [EMBL accession code: FM211187]. Alternating orange and brown boxes indicate the individual COG sequences that comprise each transport system. Solid vertical black lines divide the sequences associated with different transport systems. Dashed vertical black lines divide the sequences associated with alternative alleles of the same locus. The likely substrates of each transporter are listed in <u>S3 Table</u>. (C) Red cells indicate the Presence of the COG at the top of the column in the isolate specified by the phylogeny; blue cells indicate the COG is absent from the isolate. (PDF)

S4 Fig. Number of base substitutions imported by recombinations outside of the *cps* locus. In each boxplot, the recombinations affecting the *cps* locus are grouped according to their inferred impact on serotype. (A) The number of base substitutions outside the *cps* locus introduced by recombinations overlapping with the *cps* locus. Values of zero indicate the recombination entirely lay within the *cps* locus; values greater than zero reflect the extent to which the recombination extended into the regions flanking the *cps* locus. (B) The number of

base substitutions outside the *cps* locus introduced by recombinations occurring on the same branch of the phylogeny as a recombination overlapping with the *cps* locus. This shows the genetic diversity imported by recombinations that are likely to be contemporaneous, or nearcontemporaneous, with a recombination affecting the *cps* locus. (PDF)

S1 Table. List of serotype diversity within sequence clusters used in the permutation tests. (DOCX)

S2 Table. Linkage of serotype switches and changes in β -lactam resistance. Recombinations affecting the *cps* locus are listed in the same order as in Fig. 4, with the exception that those occurring on the same branch of the phylogeny were merged into a single row in this table. The changes in phenotype along the corresponding branch, in terms of capsule and β -lactam resistance profile (as estimated by both maximum likelihood and maximum parsimony approaches), are detailed.

(DOCX)

S3 Table. Functional characteristics of carbohydrate transporters. Each transport system is labelled with the relevant gene name, locus tag from *S. pneumoniae* TIGR4, or locus tag from *S. pneumoniae* ATCC 700669. The putative substrates are summarised from Bidossi *et al.* (DOCX)

S1 Text. Reconstruction of serotype switches.

(DOCX)

S2 Text. Detail of individual changes in β-lactam resistance. (DOCX)

Author Contributions

Conceived and designed the experiments: NJC WPH ML JP SDB JAF. Performed the experiments: CMT LK. Analyzed the data: NJC LK. Wrote the paper: NJC SDB JP WPH ML JAF.

References

- Kadioglu A, Weiser JN, Paton JC, Andrew PW (2008) The role of *Streptococcus pneumoniae* virulence factors in host respiratory colonization and disease. Nat Rev Microbiol 6: 288–301. doi: <u>10.1038/</u> <u>nrmicro1871</u> PMID: <u>18340341</u>
- Bentley SD, Aanensen DM, Mavroidi A, Saunders D, Rabbinowitsch E, et al. (2006) Genetic analysis of the capsular biosynthetic locus from all 90 pneumococcal serotypes. PLoS Genet 2: e31. PMID: <u>16532061</u>
- Park IH, Park S, Hollingshead SK, Nahm MH (2007) Genetic basis for the new pneumococcal serotype, 6C. Infect Immun 75: 4482–4489. PMID: <u>17576753</u>
- Calix JJ, Nahm MH (2010) A new pneumococcal serotype, 11E, has a variably inactivated wcjE gene. Journal of Infectious Diseases 202: 29–38. doi: <u>10.1086/653123</u> PMID: <u>20507232</u>
- Calix JJ, Porambo RJ, Brady AM, Larson TR, Yother J, et al. (2012) Biochemical, Genetic, and Serological Characterization of Two Capsule Subtypes among *Streptococcus pneumoniae* Serotype 20 Strains. Journal of Biological Chemistry 287: 27885–27894. doi: <u>10.1074/jbc.M112.380451</u> PMID: <u>22736767</u>
- Salter SJ, Hinds J, Gould KA, Lambertsen L, Hanage WP, et al. (2012) Variation at the capsule locus, cps, of mistyped and non-typable Streptococcus pneumoniae isolates. Microbiology 158: 1560–1569. doi: 10.1099/mic.0.056580-0 PMID: 22403189
- Bratcher PE, Kim KH, Kang JH, Hong JY, Nahm MH (2010) Identification of natural pneumococcal isolates expressing serotype 6D by genetic, biochemical and serological characterization. Microbiology 156: 555–560. doi: <u>10.1099/mic.0.034116-0</u> PMID: <u>19942663</u>

- Hyams C, Camberlein E, Cohen JM, Bax K, Brown JS (2010) The Streptococcus pneumoniae capsule inhibits complement activity and neutrophil phagocytosis by multiple mechanisms. Infect Immun 78: 704–715. doi: <u>10.1128/IAI.00881-09</u> PMID: <u>19948837</u>
- Lipsitch M, Whitney CG, Zell E, Kaijalainen T, Dagan R, et al. (2005) Are anticapsular antibodies the primary mechanism of protection against invasive pneumococcal disease? PLoS medicine 2: e15. PMID: 15696204
- Weinberger DM, Dagan R, Givon-Lavi N, Regev-Yochay G, Malley R, et al. (2008) Epidemiologic evidence for serotype-specific acquired immunity to pneumococcal carriage. J Infect Dis 197: 1511–1518. doi: 10.1086/587941 PMID: 18471062
- Hill PC, Cheung YB, Akisanya A, Sankareh K, Lahai G, et al. (2008) Nasopharyngeal carriage of *Streptococcus pneumoniae* in Gambian infants: a longitudinal study. Clin Infect Dis 46: 807–814. doi: <u>10.</u> <u>1086/528688</u> PMID: <u>18279039</u>
- Goldblatt D, Hussain M, Andrews N, Ashton L, Virta C, et al. (2005) Antibody responses to nasopharyngeal carriage of *Streptococcus pneumoniae* in adults: a longitudinal household study. Journal of Infectious Diseases 192: 387–393. PMID: <u>15995951</u>
- Cobey S, Lipsitch M (2012) Niche and neutral effects of acquired immunity permit coexistence of pneumococcal serotypes. Science 335: 1376–1380. doi: 10.1126/science.1215947 PMID: 22383809
- Ghaffar F, Barton T, Lozano J, Muniz LS, Hicks P, et al. (2004) Effect of the 7-valent pneumococcal conjugate vaccine on nasopharyngeal colonization by *Streptococcus pneumoniae* in the first 2 years of life. Clinical Infectious Diseases 39: 930–938. PMID: <u>15472842</u>
- Rinta-Kokko H, Dagan R, Givon-Lavi N, Auranen K (2009) Estimation of vaccine efficacy against acquisition of pneumococcal carriage. Vaccine 27: 3831–3837. doi: <u>10.1016/j.vaccine.2009.04.009</u> PMID: <u>19490983</u>
- Huang SS, Hinrichsen VL, Stevenson AE, Rifas-Shiman SL, Kleinman K, et al. (2009) Continued impact of pneumococcal conjugate vaccine on carriage in young children. Pediatrics 124: e1–11. doi: <u>10.</u> <u>1542/peds.2008-3099</u> PMID: <u>19564254</u>
- 17. Millar EV, O'Brien KL, Watt JP, Bronsdon MA, Dallas J, et al. (2006) Effect of community-wide conjugate pneumococcal vaccine use in infancy on nasopharyngeal carriage through 3 years of age: a cross-sectional study in a high-risk population. Clin Infect Dis 43: 8–15. PMID: 16758412
- Lipsitch M (1997) Vaccination against colonizing bacteria with multiple serotypes. Proc Natl Acad Sci U S A 94: 6571–6576. PMID: 9177259
- Lipsitch M (1999) Bacterial vaccines and serotype replacement: lessons from Haemophilus influenzae and prospects for Streptococcus pneumoniae. Emerg Infect Dis 5: 336–345. PMID: 10341170
- Spratt BG, Greenwood BM (2000) Prevention of pneumococcal disease by vaccination: does serotype replacement matter? Lancet 356: 1210–1211. PMID: <u>11072934</u>
- Hanage WP, Bishop CJ, Huang SS, Stevenson AE, Pelton SI, et al. (2011) Carried pneumococci in Massachusetts children: the contribution of clonal expansion and serotype switching. Pediatr Infect Dis J 30: 302–308. doi: <u>10.1097/INF.0b013e318201a154</u> PMID: <u>21085049</u>
- Robinson KA, Baughman W, Rothrock G, Barrett NL, Pass M, et al. (2001) Epidemiology of invasive Streptococcus pneumoniae infections in the United States, 1995–1998: Opportunities for prevention in the conjugate vaccine era. JAMA 285: 1729–1735. PMID: <u>11277827</u>
- Hanage WP, Finkelstein JA, Huang SS, Pelton SI, Stevenson AE, et al. (2010) Evidence that pneumococcal serotype replacement in Massachusetts following conjugate vaccination is now complete. Epidemics 2: 80–84. doi: <u>10.1016/j.epidem.2010.03.005</u> PMID: <u>21031138</u>
- Black S, Shinefield H, Fireman B, Lewis E, Ray P, et al. (2000) Efficacy, safety and immunogenicity of heptavalent pneumococcal conjugate vaccine in children. Northern California Kaiser Permanente Vaccine Study Center Group. Pediatr Infect Dis J 19: 187–195. PMID: <u>10749457</u>
- Croucher NJ, Finkelstein JA, Pelton SI, Mitchell PK, Lee GM, et al. (2013) Population genomics of postvaccine changes in pneumococcal epidemiology. Nature genetics 45: 656–663. doi: <u>10.1038/ng.2625</u> PMID: <u>23644493</u>
- 26. Wyres KL, Lambertsen LM, Croucher NJ, McGee L, von Gottberg A, et al. (2013) Pneumococcal Capsular Switching: A Historical Perspective. Journal of Infectious Diseases 207: 439–449. doi: <u>10.1093/</u> infdis/jis703 PMID: <u>23175765</u>
- 27. Feil EJ, Smith JM, Enright MC, Spratt BG (2000) Estimating recombinational parameters in *Streptococcus pneumoniae* from multilocus sequence typing data. Genetics 154: 1439–1450. PMID: <u>10747043</u>
- Croucher NJ, Harris SR, Barquist L, Parkhill J, Bentley SD (2012) A high-resolution view of genomewide pneumococcal transformation. PLoS Pathog 8: e1002745. doi: <u>10.1371/journal.ppat.1002745</u> PMID: <u>22719250</u>

- Croucher NJ, Harris SR, Fraser C, Quail MA, Burton J, et al. (2011) Rapid pneumococcal evolution in response to clinical interventions. Science 331: 430–434. doi: <u>10.1126/science.1198545</u> PMID: <u>21273480</u>
- Croucher NJ, Hanage WP, Harris SR, McGee L, van der Linden M, et al. (2014) Variable recombination dynamics during the emergence, transmission and 'disarming' of a multidrug-resistant pneumococcal clone. BMC biology 12: 49. doi: 10.1186/1741-7007-12-49 PMID: 24957517
- Hausdorff WP, Feikin DR, Klugman KP (2005) Epidemiological differences among pneumococcal serotypes. The Lancet Infectious Diseases 5: 83–93. PMID: <u>15680778</u>
- Trzcinski K, Thompson CM, Lipsitch M (2004) Single-step capsular transformation and acquisition of penicillin resistance in *Streptococcus pneumoniae*. J Bacteriol 186: 3447–3452. PMID: <u>15150231</u>
- Croucher NJ, Chewapreecha C, Hanage WP, Harris SR, McGee L, et al. (2014) Evidence for soft selective sweeps in the evolution of pneumococcal multidrug resistance and vaccine escape. Genome Biology and Evolution 6: 1589–1602. doi: <u>10.1093/gbe/evu120</u> PMID: <u>24916661</u>
- Gupta S, Maiden MC, Feavers IM, Nee S, May RM, et al. (1996) The maintenance of strain structure in populations of recombining infectious agents. Nature medicine 2: 437–442. PMID: 8597954
- Mavroidi A, Aanensen DM, Godoy D, Skovsted IC, Kaltoft MS, et al. (2007) Genetic relatedness of the Streptococcus pneumoniae capsular biosynthetic loci. Journal of bacteriology 189: 7841–7855. PMID: 17766424
- Mostowy R, Croucher NJ, Hanage WP, Harris SR, Bentley S, et al. (2014) Heterogeneity in the Frequency and Characteristics of Homologous Recombination in Pneumococcal Evolution. PLOS Genetics 10: e1004300. doi: <u>10.1371/journal.pgen.1004300</u> PMID: <u>24786281</u>
- Weinberger DM, Trzcinski K, Lu YJ, Bogaert D, Brandes A, et al. (2009) Pneumococcal capsular polysaccharide structure predicts serotype prevalence. PLoS Pathog 5: e1000476. doi: <u>10.1371/journal.</u> <u>ppat.1000476</u> PMID: <u>19521509</u>
- Bidossi A, Mulas L, Decorosi F, Colomba L, Ricci S, et al. (2012) A functional genomics approach to establish the complement of carbohydrate transporters in *Streptococcus pneumoniae*. PLoS One 7: e33320. doi: 10.1371/journal.pone.0033320 PMID: 22428019
- Tettelin H, Nelson KE, Paulsen IT, Eisen JA, Read TD, et al. (2001) Complete genome sequence of a virulent isolate of *Streptococcus pneumoniae*. Science 293: 498–506. PMID: <u>11463916</u>
- Kryazhimskiy S, Dushoff J, Bazykin GA, Plotkin JB (2011) Prevalence of epistasis in the evolution of influenza A surface proteins. PLOS Genetics 7: e1001301. doi: <u>10.1371/journal.pgen.1001301</u> PMID: <u>21390205</u>
- Croucher NJ, Coupland PG, Stevenson AE, Callendrello A, Bentley SD, et al. (2014) Diversification of bacterial genome content through distinct mechanisms over different timescales. Nature communications 5:5471. doi: <u>10.1038/ncomms6471</u> PMID: <u>25407023</u>
- Humbert O, Prudhomme M, Hakenbeck R, Dowson CG, Claverys J-P (1995) Homeologous recombination and mismatch repair during transformation in *Streptococcus pneumoniae*: saturation of the Hex mismatch repair system. Proceedings of the National Academy of Sciences 92: 9052–9056. PMID: 7568071
- Chewapreecha C, Harris SR, Croucher NJ, Turner C, Marttinen P, et al. (2014) Dense genomic sampling identifies highways of pneumococcal recombination. Nature genetics 46: 305–309. doi: <u>10.1038/</u> ng.2895 PMID: <u>24509479</u>
- 44. Johnston C, Martin B, Granadel C, Polard P, Claverys J-P (2013) Programmed protection of foreign DNA from restriction allows pathogenicity island exchange during pneumococcal transformation. PLoS pathogens 9: e1003178. doi: 10.1371/journal.ppat.1003178 PMID: 23459610
- 45. Hathaway LJ, Brugger SD, Morand B, Bangert M, Rotzetter JU, et al. (2012) Capsule type of Streptococcus pneumoniae determines growth phenotype. PLoS pathogens 8: e1002574. doi: <u>10.1371/</u> journal.ppat.1002574 PMID: <u>22412375</u>
- 46. Croucher NJ, Page AJ, Connor TR, Delaney AJ, Keane JA, et al. (2014) Rapid phylogenetic analysis of large samples of recombinant bacterial whole genome sequences using Gubbins. Nucleic acids research: gku1196.
- Paradis E, Claude J, Strimmer K (2004) APE: analyses of phylogenetics and evolution in R language. Bioinformatics 20: 289–290. PMID: <u>14734327</u>
- Sung C, Li H, Claverys J, Morrison D (2001) An *rpsL* cassette, Janus, for gene replacement through negative selection in *Streptococcus pneumoniae*. Applied and environmental microbiology 67: 5190–5196. PMID: <u>11679344</u>
- Trzcinski K, Thompson CM, Lipsitch M (2003) Construction of otherwise isogenic serotype 6B, 7F, 14, and 19F capsular variants of *Streptococcus pneumoniae* strain TIGR4. Applied and environmental microbiology 69: 7364–7370. PMID: <u>14660386</u>